Author's response to reviews

Title: Perinatal factors associated with early neonatal deaths in very low birth weight preterm infants in Northeast Brazil

Authors:

Eveline CM Castro (evelinecamposmc@gmail.com)
Alvaro JM Leite (alvaromadeiro@yahoo.com.br)
Maria FB Almeida (fernandaalmeida@uol.com.br)
Ruth Guinsburg (ruthgbr@netpoint.com.br)

Version: 5 Date: 16 November 2014

Author's response to reviews:

São Paulo, November 16, 2014.

To
Peter O'Donovan,
Executive Editor, BMC Pediatrics

Dear Dr. O'Donovan,

Thank you again for the review of our manuscript. Besides the changes made in response to the reviewers and sent last October 16, 2014, minor changes asked by the Handling Editor were done: the English grammar and spelling was reviewed again by American Journal Experts (1st and 2nd certificates enclosed) and the whole text and the title of the figures were updated to reflect the suggested wording of the Reviewers (0-6 days, rather than 168 hours)

Ruth Guinsburg (ruthgbr@netpoint.com.br) and Eveline Campos Monteiro de Castro (evelinecamposmc@gmail.com), on behalf of all authors